MATH 220 - Spring 1999 Matrices

http://www.math.psu.edu/alm/math220.html TR - 10:10 - 11:00, 217 Willard Instructor: Alexandre L. Madureira, alm@math.psu.edu Office: 432 McAllister Building, 865-4291

Office hours:

• Tuesday 11:00–12:00 • Friday 2:30–3:30

Grades:

Homework (50 points) – I’ll assign homework (almost) every lecture and you’ll have to submit it in the following one. I’ll ask the grader to correct few selected questions.

Quizzes (50 points) – There’ll be quizzes every two weeks, and I’ll let you know in advance.